

Article

Correlated Functional Models with Derivative Information for Modeling Microfading Spectrometry Data on Rock Art Paintings

Gabriel Riutort-Mayol ^{1,*}, Virgilio Gómez-Rubio ², José Luis Lerma ¹, and Julio M. del Hoyo-Meléndez ³

- ¹ Department of Cartographic Engineering, Geodesy, and Photogrammetry, Universitat Politècnica de València, 46022 Valencia, Spain; jllerma@cgf.upv.es
- ² Department of Mathematics, School of Industrial Engineering, Universidad de Castilla-La Mancha, 02071 Albacete, Spain; Virgilio.Gomez@uclm.es
- ³ Laboratory of Analysis and Non-Destructive Investigation of Heritage Objects, The National Museum in Krakow, 30-062 Krakow, Poland; jdelhoyo@mnk.pl
- * Correspondence: gabriuma@gmail.com

Received: 16 October 2020; Accepted: 24 November 2020; Published: 1 December 2020



Abstract: Rock art paintings present high sensitivity to light, and an exhaustive evaluation of the potential color degradation effects is essential for further conservation and preservation actions on these rock art systems. Microfading spectrometry (MFS) is a technique that provides time series of stochastic observations that represent color fading over time at the measured points on the surface under study. In this work, a reliable and robust modeling framework for a short and greatly fluctuating observation dataset collected over the surfaces of rock art paintings located on the walls of Cova Remigia in Ares del Maestrat, Castellón, Spain, is presented. The model is based on a spatially correlated spline-based time series model that takes into account prior information in the form of model derivatives to guarantee monotonicity and long-term saturation for predictions of new color fading estimates at unobserved locations on the surface. The correlation among the (spatially located) time series is modeled by defining Gaussian process (GP) priors over the spline coefficients across time series. The goal is to obtain a complete spatio-temporal mapping of color fading estimates for the study area, which results in very important and useful information that will potentially serve to create better policies and guidelines for heritage preservation and sustainable rock art cultural tourism.

Keywords: spatio-temporal statistics; penalized splines; Gaussian processes; derivative constraining observations; monotonicity; microfade testing

1. Introduction

Prehistoric rock art paintings (see, for example, Figure 1, left plot) are exposed to environmental elements that can accelerate their degradation, increasing the risk of losing such valuable cultural assets from past societies. The study and documentation of potential damages to these rock art systems are essential for their conservation and preservation [1–3], and it is a priority for cultural and research organizations, scientists and government agencies. The present research is focused on the paintings located in Cova Remigia open-air rock art shelter in Castellón (Spain), which is part of the Spanish Levantine rock art paintings of the Mediterranean basin on the Iberian Peninsula declared World Heritage by UNESCO in 1998. There is still a lack of knowledge about their conservation status and how rock art is deteriorating [4–6].





Figure 1. (a) View of Cova Remigia in Ares del Maestrat, Castellón, Spain. (b) Placement of the MFS instrument for one measurement.

Rock art paintings are prone to different types of weathering, not only due to human activity but mainly through natural weathering processes [7,8]. Particularly, paintings of Levantine environments such as Cova Remigia are at risk, threatened by warming temperatures, higher seasonal variable precipitation and increased wind speeds. Furthermore, long and intense exposure to sunlight has adverse effects on these open-air shelters due to thermal and photochemical degradation of the historic materials, and changes in the spectral properties (color) of the materials are one of the main effects. Heating from insolation results in rapid temperature variations on the rock surface relative to its interior causing physical failure. In addition, photochemical degradation of pigments has also been observed. This is a surface effect, and the degree of color change depends on the chemical composition of the pigment [4,9].

This study is focused on the study and documentation of the degree of color-changing/fading of paintings, patinas and host rock due to the effects of insolation. Materials with higher light sensitivity usually experience rapid color changes during the early stages of light exposure, followed by slower rates after maximum fading has occurred; the total disappearance of the substance that produces the color (chromophore) occurs at this second stage of the fading [10–13]. For these reasons, fading cannot decrease with time and it is expected to stabilize in the long term. Materials can show different times to saturation depending on their physicochemical properties and concentrations of chromophores [3].

Microfading spectrometry (MFS) is a method for assessing the stability to light irradiation of cultural heritage objects. This method allows for real-time monitoring of spectral changes of a cultural heritage object by undertaking in situ lightfastness measurements on the surface under study [14,15]. MFS provides for each measured point on the surface of a rock art painting a time series of observations that represents potential color fading over time given exposure to light [16]. Hence, microfading measurements can be seen as observations of an underlying spatio-temporal stochastic process.

The microfading instrument is very sensitive to movement and glossy surface effects, causing extremely large fluctuations in the measurements. Furthermore, they can be easily contaminated by changes in the lighting conditions when the measurements are performed in outside environments. These large fluctuations and possible systematic noise effects in the observations can lead to models that do not satisfy those properties of non-decreasing monotonicity and long-term stabilization of color fading over time. Thus, it is recommended to include additional information in the models in order to meet these properties and to ensure reliable properties for color fading estimates in new unobserved locations. Existing lightfastness studies on these systems have been limited to analyzing a few measured points on the surfaces of the rock art paintings because these measurements are largely time-consuming and difficult to materialize under harsh conditions (Figure 1, right plot).

Hence, in this paper a reliable modeling framework for a short and greatly fluctuating observation dataset is proposed, aiming at extending the analysis of color fading at locations on the surfaces of

rock art paintings where no observations are available. The model is based on spatially correlated time series functions with the consideration of additional information on the derivatives of the functions aiming at ensuring the functions are monotonically non-decreasing and saturated in the long term. A complete map of potential color fading estimates for the entire surface under study and over time offers important and useful information in order to achieve further successful conservation actions and policies. Although color fading is only one of the potentially damaging effects on the surfaces of rock art paintings, color fading is an important effect to control.

Information of potential color fading estimates, alone or in conjunction with the evaluation of other potential damaging factors, will help to develop conservation policies. A comprehensive evaluation of the state of preservation of rock art paintings together with a risk assessment plan that considers their potential damage due to exposure to environmental elements will serve to better allocate resources and devise a plan of heritage conservation tasks for these rock art systems. The research will also allow creating policies and guidelines for sustainable rock art cultural tourism, guaranteeing the conservation and preservation of this enormous patrimonial legacy for the enjoyment of present and future generations and at the same time creating further economic benefits in the region and community. Once the key elements that affect the long-term conservation of the rock art paintings are well understood, the next step is to consider the economic factors that drive tourism and increase the public interest in this geographical area. This is an enormous challenge, as rock art is of evident contemporary cultural and social value, but at the same time it can be difficult to create sustainable tourism businesses centered around these sites. This makes good understanding and documentation of the potential damages on these systems an essential matter.

The rest of the paper is structured as follows. In Section 2, a brief bibliographic review of correlated functional models and monotonic approximations is done. Section 3 states the proposed modeling methodology in detail. Section 4 describes the case study and the available data set. Sections 5 and 6 focus on the modeling and inference formulation, respectively. Section 7 describes model checking and model selection procedures. Section 8 analyzes and evaluates the fit of the model and describes the results of applying the model to the data set. Section 9 briefly discusses the conducted research and results in relation to the applied field. Section 10 presents brief conclusions of the work.

2. Background Methods

Functional data usually refer to independent realizations of a functional random variable that take values in a continuous space. Time series of observations (e.g., color-fading time series) might be the most common case of functional data in 1D, but spatially distributed observations can also be seen as functional data in a 2D space and spatio-temporal observations in a 3D space. In order to construct a model useful for making predictions of new functional data as a function of new values of the variables in the input space, the process must be considered as an structured process with dependence among observations.

Correlated functional models consider the observed functional data as non-independent functions. A popular approach for correlated space-time functional data consists of three-way (spatial, 2D, and temporal, 1D) penalized spline models [17] with different basis constructions based on Kronecker products [18,19] or additive basis components [20]. Aguilera, et al. [21] proposed a mixture of a functional regression model for functional response and a penalized spline spatial regression. However, in general, these models have large numbers of parameters.

Another interesting approach consists of considering the space-time structured observations as stochastic realizations of a Gaussian process (GP) prior with a spatio-temporal covariance function. GPs [22,23] are natural and flexible non-parametric prior models for *D*-dimensional (e.g., space and time) functions with multivariate predictors (input variables) in each dimension. In a separable form, the space-time covariance function is made of two independent processes, space and time [24]. In a non-separable form, the covariance function models space-time interaction [25,26]. The drawback of GP models is their expensive computational demand in covariance matrix inversion that is in general

a $O((NT)^3)$ operation, with *N* being the number of spatial locations, with coordinates (S_x, S_y) , and *T* being the number of time points.

A geostatistical approach for spatio-temporal data is the kriging approach for 1D-functional data [27], in which the 1D-functional data are a time series. The spatial correlation of the time series is modeled in the covariance function. The dimension of the covariance matrix is $N \times N$ and the matrix inversion is an $O(N^3)$ operation. Although this approach requires less computational effort than the spatio-temporal GPs, it has the drawback of having less flexible spatio-temporal structure since the same spatial structure is defined for all time series. A related approach can be found in [28], where the spatial correlation between the time series is modeled by defining GPs with a spatial covariance function across the time series functional coefficients. This construction allows for modeling different spatial structures for the different orders of the coefficients.

Inducing monotonicity or gradient to functions can be expressed in terms of the derivative of the function, as either the sign of a derivative for monotonicity or the value of a derivative for gradients. As differentiation is a linear operator, the derivatives of both semiparametric models and GP models can be used as additional observations jointly with the regular observations in order to force the function to fit these properties [29,30]. Riihimäki and Vehtari [31] used a GP model and the information of its derivative process to induce monotonicity on the functions using virtual observations of the sign of the derivatives. It is known that there are practical issues with this approach of inducing monotonicity through virtual observations, since the posterior distributions of the functions depend on the number of the inducing points; when the number of points is too large, the posteriors tend to be overly smoothed. This over-smoothing effect appears to be particularly significant when using GPs as compared to splines. However, if the function is smooth, this problem can be avoided in practice by choosing only a few inducing points ([32], Chapter 4). In [33] the authors used the same idea of using virtual observations to include constraints concerning the derivative of the functions in deep probabilistic models.

On the other hand, monotonicity in semiparametric models can be forced by construction. Shively, et al. [30] proposes two approaches to obtain monotonic functions, the first using a modified characterization of the smooth monotone functions proposed in [34] that allows for unconstrained estimation, and the second using constrained prior distributions for the regression coefficients to ensure monotonicity. Brezger and Steiner [35] induces monotonicity on semiparametric and non-parametric models by imposing the restriction that the regression coefficients are ordered, sometimes also known as isotonic regression. These constraints are imposed by specifying truncated prior distributions in order to reject the undesired draws for the parameters in the MCMC sampling. Reich, et al. [36] used a similar approach: imposing order to the regression parameters by means of reparameterizing and constraining these parameters with application to a quantile regression model. Regarding imposing monotonicity by construction in non-parametric GP prior models, this becomes more difficult. Generic GP prior models do not restrict the function values to be monotonically increasing or decreasing with respect to input variables. Recently, Andersen, et al. [37] proposed a monotonic model based on applying a non-linear transformation of a GP and then using Hilbert space methods to approximate the model to make the inference tractable [38].

3. Proposed Methodology

In this study, a specific application with the aim of modeling and predicting color fading time series at new unobserved spatial locations on the surfaces of rock art paintings was carried out. The main motivation of this study is to construct a model that exploits the correlation structure of the data in order to extend the analysis and make useful predictions, in the scenario of a short set of sampling observations, as in the case of microfading measurements on rock art paintings. In fact, in the present csae study, only 13 measured locations on the surface were collected. A spatially correlated time series model with information of the derivative process is proposed, and the spatial correlation among time series and their derivatives is established by defining multivariate GPs over the spline

5 of 25

coefficients across the time series. A GP is a natural way to evaluate the covariance structure of the data and use it to make useful generalizations. This approach of combining spline functions for the temporal dimension and multivariate GPs to spatially correlate the spline coefficients is an original spatio-temporal approach, seldom implemented in practice (a related but simpler approach can be found in [28]) that, as discussed in Section 6, is flexible and can reasonably consider different covariance structures, with different complexities between space and knots.

The color-fading time series are modeled as penalized cubic-spline functions. The derivative of a semiparametric model such as a spline model is still a linear model and can be used as an additional observation together with actual observations, thereby encoding derivative information of the functions into the modeling. Derivative observations of both the sign and the values of the partial derivatives are used to induce (non-decreasing) monotonicity and long-term saturation of color fading (long term color fading [3]), respectively, as a function of time. In order to force the functions to be zero at the starting points of every time series, observations with the absence of noise are used at these points. A thorough illustration of the implementation of both zero and first order modeling constraints to control the values of the function and the dynamics (monotonicity and gradients) of the function, respectively, in a full Bayesian framework, was carried out. Monotonicity through virtual observations in Bayesian GPs was originally proposed by Riihimäki and Vehtari [31], and in this work, this idea is used on spline functions.

Color fading is mainly related to physicochemical properties of the measured surface. Physicochemical data for all the points on the surface as inputs to predict new curves are hard to obtain. Instead, image color values can be used as surrogate input variables to construct and evaluate the correlation, since these image color variables are known to be related to the physicochemical properties of the imaged material [39]. A multivariate covariance function in a GP allows modeling trichromatic image color variables jointly with spatial distances as inputs to evaluate the covariance structure of the data.

For model evaluation and comparison, the same model but without derivative information was also fitted. Cross-validation procedures were conducted (see Section 7) to compute the posterior predictive checks, the expected log predictive density and the mean square predictive errors in order to do model checking and selection and evaluate the predictive performance.

4. Case Study and Data Description

Microfade testing allows for real-time monitoring of spectral changes of a small area (diameter of 0.5 mm) of a cultural heritage object by exposing it to light. This method was developed by Whitmore and others [14,15] and has been extensively used to study the lightfastness of cultural objects [40–42]. In this practical case, we sought to evaluate and document the degree of color fading over time and space on rock art paintings located in Cova Remigia rock-shelter (in Ares del Maestrat, Castellón, Spain) caused by direct solar irradiation. This site is particularly interesting because there are hundreds of tiny (cm-level) rock art figures. These have been drawn using black and red pigments (and white), instead of red like the rest of rock art caves, which makes Cova Remigia particularly important in the overall context. There is still a lack of knowledge about their conservation status and how the rock art is deteriorating. The current study analyzes photostability and potential color change, which has important implications for Cova Remigia, since it describes the paintings and explains their long-term degradation under future environmental conditions.

MFS testing at each measured point on the surface gives rise to a time series that represents color fading over time. Color fading can be described in terms of CIE color differences ΔE_{ab}^* [39]. The microfading measurements have a duration of 10 min. The sampling frequency will be once per minute, such that the resulting time series will consist of 11 time points (T = 11), represented by the time input variable $t = \{t_l = l : l = 1, 2, ..., T\}$. Since these measurements are largely time-consuming and difficult to materialize, especially in these rock art systems, the MFS dataset available only consisted of 13 observed locations on the surface (N = 13). Figure 2-left shows their pixel locations on

a color image of the study area; each location incorporates color fading time series of observations (Figure 2, right plot).



Figure 2. (a) Image with the spatial locations where the MFS observations were measured in Shelter V of Cova Remigia. (b) Observed MFS time series at every sampled spatial location on the surface.

The number of available input variables to evaluate the covariance between spatial locations is 5 (D = 5), which are the three image color variables per spatial location i (i = 1, 2, ..., N): hue (H_i), saturation (S_i) and intensity (I_i), and the two spatial coordinates S_{xi} and S_{yi} . Table 1 contains summary statistics for these input variables, which have been rescaled by dividing by their standard deviations. Spatial coordinates S_x and S_y were divided by their common standard deviation.

	Н	S	Ι	S_x	S_y
Mean	5.255	9.704	5.155	3.549	4.969
Std. Dev.	1.0	1.0	1.0	0.732	0.674
1st Qu.	4.359	9.042	4.603	2.910	4.387
3rd Qu.	5.913	10.273	5.772	4.187	5.551

Table 1. Summary statistics of the input variables.

Each time series is modeled as a spline-based function. The order or number of knots *K* of the spline functions is set to 3 and placed uniformly through the time points variable. Due to the large fluctuations present in the measurements conducted on these rock art systems and the limited sample data set, some modeling issues can arise, such as not starting at zero, not being monotonically non-decreasing or not stabilizing in the long term as a function of time, which are properties assumed for the color fading curves, as discussed above. Most of the rock art painting systems analyzed so far show stabilization at or before the 10-minute MFS monitoring measurements.

Therefore, the functions must be constrained to be zero at observations in subset $B = \{(l, i) : l = 1\}$ of starting points of every time series, that is, the predicted time series must start at zero. Subset $E = \{(l, i) : l = 11\}$ contains the ending points of every time series, the functions will be constrained to reach a stabilization as a function of time. The functions are guaranteed to be monotonically non-decreasing as a function of time when the partial derivative is non-negative. Virtual points for monotonicity will be appropriately placed at two time points of every time series, denoted by subset $F = \{(l, i) : l \in \{7, 10\}\}$ of observation indices. The use of only two virtual points in every time series will probably be enough to ensure monotonicity in expectation for the entire time series, since the time series functions to be learned are expected to be smooth. Additionally, only two virtual points to prevent posterior functions from being overly smoothed are used, as commented on in the previous Section 2, even though this is known not to affect splines severely ([32], Chapter 4).

The actual equivalency of the exposure time used in MFS in years depends on the hours and intensity of sunlight that affects the paintings on a changing daily basis. Without proper monitoring of light, this equivalency is difficult to obtain, so this aspect of the research is not considered in the current study.

5. Spatially Correlated Time-Series Model with Derivative Information

5.1. Observational Model

We consider a continuous stochastic process based on time series observed at spatial locations on a surface. Thus, y_{li} denotes the *l*-th value of the time series at the *i*th spatial location, with l = 1, 2, ..., T representing the time points and i = 1, 2, ..., N representing the spatial locations. We adopt the model $y_{li} = f_{li} + \epsilon_{li}$, where f_{li} is the value of the latent function $f_i : \mathbb{R} \to \mathbb{R}$ evaluated at the time input value t_l , i.e., $f_{li} = f_i(t_l)$, and $\epsilon_{li} \sim \mathcal{N}(0, \sigma^2)$ is the Gaussian noise. The latent function $f_i(\cdot)$ underlies the time series of noisy observations for the *i*th spatial location. Let $y \in \mathbb{R}^{T \times N}$ and $f \in \mathbb{R}^{T \times N}$ be the matrices of spatio-temporal noisy observations and latent function values, respectively.

5.2. Penalized Spline Time-Series Functions

We consider a penalized thin-plate cubic-spline function for modeling the latent function of each time series of the spatio-temporal data. It is worth noting that quadratic-splines may be sufficient as the observed color fading curves seem quite smooth. However, cubic-splines have been implemented in this paper for generality and modeling flexibility. In particular, penalized thin-plate splines as presented by Crainiceanu, et al. [43] have been used, which are semiparametric regression models [17,44].

Firstly, it is assumed that latent function $f_i(\cdot)$ is a thin-plate cubic-spline function:

$$f_i(t_l) = \beta_{1i} + \beta_{2i}t_l + \sum_{k=1}^{K} Z_{lk}u_{ki} = H_l \cdot \beta_{\cdot i} + Z_l \cdot u_{\cdot i},$$
(1)

where $H_{l.} = (1, t_l)$ is the row-vector of linear function values of the *l*th time point, $\beta_{.i} = (\beta_{1i}, \beta_{2i})^{\top}$ is the column-vector of linear coefficients for the *i*th time series, $Z_{l.} = (Z_{l1}, ..., Z_{lK}) \in \mathbb{R}^{K}$ is the row-vector of cubic-spline function values of the *l*th time point, and $u_{.i} = (u_{1i}, ..., u_{Ki})^{\top} \in \mathbb{R}^{K}$ is the column-vector of spline coefficients for the *i*th time series, with *K* the order of the spline function and number of knots. The *k*th element of the vector $Z_{l.}$ of cubic-spline function values is

$$Z_{lk} = \phi(t_l, \kappa_k) = \frac{1}{4} \left((\kappa_k - 0.5)^2 - \frac{1}{12} \right) \left((t_l - 0.5)^2 - \frac{1}{12} \right) - \frac{1}{24} \left((|t_l - \kappa_k| - 0.5)^4 - \frac{1}{2} (|t_l - \kappa_k| - 0.5)^2 \right),$$
(2)

which is the one-dimensional cubic-spline function ϕ [45] evaluated at the *l*th time point and the pre-fixed knot κ_k corresponding to the *k*th spline component, with k = 1, 2, ..., K.

Next, the penalized representation of the thin-plate spline function in Equation (1) is derived. A more detailed explanation of this approach can be found in Crainiceanu, et al. [43]. In order to derive the penalized representation, let first formulate the complete latent model for the time series of observations at spatial location *i*:

$$\mathbf{y}_{\cdot i} = f_{\cdot i} + \boldsymbol{\epsilon}_{\cdot i} = H\boldsymbol{\beta}_{\cdot i} + Z\boldsymbol{u}_{\cdot i} + \boldsymbol{\epsilon}_{\cdot i},$$

where $\mathbf{y}_{.i} = (y_{1i}, \ldots, y_{Ti})^{\top} \in \mathbb{R}^T$ and $f_{.i} = (f_{1i}, \ldots, f_{Ti})^{\top} \in \mathbb{R}^T$ are the column-vectors of noisy observations and latent function values, respectively, for spatial location *i*, where latent function value f_{li} is the value of the latent function $f_i(\cdot)$ evaluated at the time input value t_l , i.e., $f_{li} = f_i(t_l)$. $H \in \mathbb{R}^{T \times 2}$ denotes the matrix of linear function values with *l*-th row $H_{l.}, Z \in \mathbb{R}^{T \times K}$ denotes the matrix of cubic-spline function values with *l*th row $Z_{l.}$, and $\epsilon_{i} \sim \mathcal{N}(0, \sigma^2)$ is the Gaussian noise term. To perform the penalization of the function in order to avoid overfitting at location *i*, the following term can be minimized:

$$\frac{1}{\sigma^2}||\boldsymbol{y}_{\cdot i} - H\boldsymbol{\beta}_{\cdot i} - Z\boldsymbol{u}_{\cdot i}||^2 + \frac{1}{\sigma_u^2}\boldsymbol{u}_{\cdot i}^\top \boldsymbol{\Omega}\boldsymbol{u}_{\cdot i}$$

in which the covariance of vector u_{i} is $cov(u_{i}) = \sigma_{u}^{2}\Omega^{-1}$, the covariance of the residuals ϵ_{i} is $cov(\epsilon_{i}) = \sigma^{2}I$ (with *I* the identity matrix), and $\Omega \in \mathbb{R}^{K \times K}$ is the matrix of penalization with an element (q, k):

$$\Omega_{qk} = \phi(\kappa_q, \kappa_k) = \frac{1}{4} \left((\kappa_k - 0.5)^2 - \frac{1}{12} \right) \left((\kappa_q - 0.5)^2 - \frac{1}{12} \right) - \frac{1}{24} \left(\left(|\kappa_q - \kappa_k| - 0.5 \right)^4 - \frac{1}{2} \left(|\kappa_q - \kappa_k| - 0.5 \right)^2 \right), \quad (3)$$

that is, the cubic-spline function ϕ evaluated at every pair of the pre-fixed knots κ_q and κ_k , corresponding to the *q*th and *k*th spline component, respectively, with q, k = 1, ..., K.

When reparametrization $u_{i} = \Omega^{-1/2} b_{i}$ is used, an equivalent model representation of the penalized thin-plate splines is obtained in the form of linear mixed models [46]:

$$\mathbf{y}_{\cdot i} = f_{\cdot i} + \boldsymbol{\epsilon}_{\cdot i} = H\boldsymbol{\beta}_{\cdot i} + Z\Omega^{-1/2}\boldsymbol{b}_{\cdot i} + \boldsymbol{\epsilon}_{\cdot i},$$

where the covariance of the vector of spline coefficients $\boldsymbol{b}_{i} \in \mathbb{R}^{K}$ is now $\operatorname{cov}(\boldsymbol{b}_{i}) = \sigma_{b}^{2}I$.

As a result, the latent function $f_i(\cdot)$ can be represented as penalized thin-plate cubic splines in the form of linear mixed models:

$$f_i(t_l) = \boldsymbol{H}_l.\boldsymbol{\beta}_{.i} + \boldsymbol{Z}_l.\boldsymbol{\Omega}^{-1/2}\boldsymbol{b}_{.i},\tag{4}$$

where $\boldsymbol{b}_{i} = (b_{1i}, \dots, b_{Ki})^{\top} \in \mathbb{R}^{K}$ is the column-vector of penalized spline coefficients for the *i*th time series.

In practice, the inverse of the square root of the matrix Ω is often computed as $\Omega^{-1/2} = UD^{-1/2}V^{\top}$, where UDV^{\top} is the singular value decomposition of the matrix Ω , where *D* is the rectangular diagonal matrix with the singular values, *U* and *V* are the matrices with the left-singular vectors and the right-singular vectors of Ω , respectively.

The matrix $\tilde{f} = [\tilde{f}_{\cdot 1} \ f_{\cdot 2} \ \cdots \ f_{\cdot N}] \in \mathbb{R}^{T \times N}$ of spatio-temporal latent function values can be expressed as:

$$f = H\boldsymbol{\beta} + Z\Omega^{-\frac{1}{2}}\boldsymbol{b},$$

where $\boldsymbol{\beta} \in \mathbb{R}^{2 \times N}$ is the matrix of linear coefficients with *i*th column $\boldsymbol{\beta}_{.i}$, $\boldsymbol{b} = [\boldsymbol{b}_{.1} \, \boldsymbol{b}_{.2} \cdots \boldsymbol{b}_{.N}] \in \mathbb{R}^{K \times N}$ is the matrix of spline coefficients with *i*th column $\boldsymbol{b}_{.i}$.

5.3. Spatially Correlating the Penalized Spline Time-Series Functions

In order to establish a correlation structure among time series, the matrix of spline coefficients $b \in \mathbb{R}^{K \times N}$ is considered as a realization of a continuous stochastic process, and a GP prior model with a two-dimensional (with *K* knots and *N* space points) covariance function can be defined on them.

To simplify the covariance structure and the computational requirements of inverting the Gram matrix of the covariance function when performing inference on GPs, a null covariance between the spline coefficients associated with a different order in the model can be considered, which is equivalent to defining *K* independent GP prior distributions, one for each *k*th row $\mathbf{b}_{k.} = (b_{k1}, \ldots, b_{kN}) \in \mathbb{R}^N$ of matrix \mathbf{b} , such that

$$\boldsymbol{b}_{k} \sim \mathcal{GP}(0, \alpha_k C_k(\boldsymbol{x}, \boldsymbol{\rho}_k)),$$

for k = 1, 2, ..., K. C_k is the covariance function for vector of coefficients \boldsymbol{b}_k , which is a function of matrix of inputs $\boldsymbol{x} = [\boldsymbol{x}_{.1} \ \boldsymbol{x}_{.2} \ \cdots \ \boldsymbol{x}_{.N}] \in \mathbb{R}^{5 \times N}$, with *i*th column $\boldsymbol{x}_{.i} = (H_i, S_i, I_i, S_{x_i}, S_{y_i})^\top \in \mathbb{R}^5$ containing the input values for the *i*th spatial location as described in Section 4, and vector of length-scale parameters $\boldsymbol{\rho}_k = (\rho_{1k}, \dots, \rho_{5k}) \in \mathbb{R}^5$ for vector \boldsymbol{b}_k . Hyperparameter α_k is the marginal variance of the GP which controls the overall scale or magnitude of the range of values of vector \boldsymbol{b}_k . Thus, specific covariance structure for each *k*th vector of spline coefficients \boldsymbol{b}_k . can be specified.

Furthermore, in this work the covariance structure is simplified even more by considering the same prior spatial structure for every vector of spline coefficients $b_{k,i}$ i.e.,

$$\boldsymbol{b}_{k.} \sim \mathcal{GP}(0, \alpha_k C(\boldsymbol{x}, \boldsymbol{\rho})),$$

and *C* is a common covariance function for every vector of coefficients \boldsymbol{b}_k , which is a function of matrix of inputs \boldsymbol{x} and length-scales $\boldsymbol{\rho} = (\rho_1, \dots, \rho_5)$. Saying that \boldsymbol{b}_k . follows a GP prior distribution of zero mean (a generic mean function can also be used) and covariance function *C* and marginal variance α_k , is equivalent to saying that \boldsymbol{b}_k . is normally distributed with zero mean and covariance matrix $\Sigma_k \in \mathbb{R}^{N \times N}$,

$$\boldsymbol{b}_{k} \sim \mathcal{N}(\boldsymbol{0}, \boldsymbol{\Sigma}_{k}), \tag{5}$$

where an element (i, j) of Σ_k is given by

$$\Sigma_{k_{(i,i)}} = \alpha_k C(\mathbf{x}_{\cdot i}, \mathbf{x}_{\cdot j}, \boldsymbol{\rho})$$

with i, j = 1, 2, ..., N, and $C(\mathbf{x}_{.i}, \mathbf{x}_{.j}, \boldsymbol{\rho})$ represents the covariance function evaluated at the specific *i*-th and *j*-th spatial locations. A squared exponential covariance function (see Rasmussen and Williams [23] for a review of covariance functions for GPs) is considered:

$$C(\mathbf{x}_{.i}, \mathbf{x}_{.j}, \boldsymbol{\rho}) = \exp\left(-\frac{1}{2}\sum_{d=1}^{5}\frac{1}{\rho_d^2}(\mathbf{x}_{di} - \mathbf{x}_{dj})^2\right),$$

where x_{di} and x_{dj} are the values of the *d*-th input variable at the *i*-th and *j*-th spatial locations, respectively. Length-scale parameters ρ_1 , ρ_2 and ρ_3 correspond to the input variables H, S and I, respectively. The spatial coordinates input variables S_x and S_y are sharing the length-scale parameter ($\rho_4 = \rho_5$), so that the covariance function depends on the (Euclidean) distance between spatial coordinates. Hyperparameter ρ_d controls the smoothness of the function for b_k in the dimension of the *d*-th input variable. The use of a multidimensional length-scale basically makes the isotropic covariance function non-isotropic. The squared exponential covariance function is a stationary function with respect to the input variables. Stationarity in covariance is a reasonable assumption for the current application. In applications where non-stationarity is desirable, valid non-stationary covariance functions for GPs can also be used (see [23] for a complete review of covariance functions for GPs).

5.4. Derivative of the Penalized Spline Time-Series Functions

The derivative function $f'_i : \mathbb{R} \to \mathbb{R}$ of penalized cubic-spline function $f_i(\cdot)$ in Equation (4) with respect to the time input variable takes the form:

$$f_i'(t_l) = \frac{\partial f_i(t_l)}{\partial t_l} = \boldsymbol{\beta}_{2i} + \frac{\partial \boldsymbol{Z}_l}{\partial t_l} \Omega^{-\frac{1}{2}} \boldsymbol{b}_{.i}, \tag{6}$$

where $\frac{\partial Z_{l.}}{\partial t_{l}} = (\frac{\partial Z_{l1}}{\partial t_{l}}, \dots, \frac{\partial Z_{lK}}{\partial t_{l}}) \in \mathbb{R}^{K}$ is the row-vector of derivative cubic-spline values at time t_{l} , and an element $\frac{\partial Z_{lk}}{\partial t_{l}}$ is the partial derivative of the cubic spline function Z_{lk} in Equation (2) with respect to the time input variable:

$$\frac{\partial Z_{lk}}{\partial t_l} = \frac{\phi(t_l, \kappa_k)}{\partial t_l} = \frac{1}{4} \left((\kappa_k - 0.5)^2 - \frac{1}{12} \right) 2(t_l - 0.5) - \frac{1}{24} \left(4 \left(|t_l - \kappa_k| - 0.5 \right)^3 - \left(|t_l - \kappa_k| - 0.5 \right) \right).$$
(7)

5.5. A Zero-Order Constraint for Time-Series to Start at Zero

In order to force predicted time series functions to start at zero, a zero-order constraint can be specified by using a Dirac Delta function $\delta(\cdot)$ as the observational model for the observations in subset $B = \{(l, i) : l = 1\}$ of starting points of every time series:

$$p(\mathbf{y}_B | \mathbf{f}_B) = \prod_{(l,i) \in B} \delta(y_{li} - f_i(t_l)),$$
$$\mathbf{y}_B = \{\mathbf{0} : (l,i) \in B\}$$
(8)

with

denoting the subset of observations in subset *B* of points. The rest of observations, i.e., observations y in subset $A = \{(l, i) : l > 1\}$ of points and denoted by

$$y_A = \{ y_{li} : (l,i) \in A \}, \tag{9}$$

are considered to be contaminated with Gaussian noise so that:

$$p(\boldsymbol{y}_A|\boldsymbol{f}_A,\sigma^2) = \prod_{(l,i)\in A} \mathcal{N}(y_{li}|f_i(t_l),\sigma^2).$$

In the previous equations, f_B and f_A denote the function values $f_i(t_l)$ in Equation (4) in subsets *B* and *A* of points, respectively.

5.6. A First-Order Derivative Constraint for Time-Series

In order to impose a saturation constraint for long-term stabilization of the time series, a set of virtual observations m_{li} of the value of the partial derivative of the function with respect to the time input variable ($m_{li} = \frac{\partial f_{li}}{\partial t_l}$) equal to zero can be considered:

$$m = \{0 : (l, i) \in E\},\tag{10}$$

where $E = \{(l, i) : l = 11\}$ is the subset of ending points of every time series to induce saturation of the function (as explained in Section 4). The Dirac Delta function $\delta(\cdot)$ can also be used as an observational model for these observations,

$$p(\boldsymbol{m}|\boldsymbol{f}_E') = \prod_{(l,i)\in E} \delta(m_{li} - f_i'(t_l)),$$

where f'_E denotes the partial derivative function values $f'_i(t_l)$ in Equation (6) in the subset *E* of points.

5.7. Monotonicity Constraint for the Time-Series

The function is guaranteed to be non-decreasing as a function of time when the partial derivative is non-negative. This constraint can be specified by using a set of virtual observations z_{li} of the sign of the partial derivative of the function with respect to the time input variable $(z_{li} = \text{sign}(\frac{\partial f_{li}}{\partial t_l}))$ equal to one:

$$z = \{1 : (l, i) \in F\},\tag{11}$$

where $F = \{(l, i) : l = \{7, 10\}\}$ is the subset of desired points where to induce monotonicity (as explained in Section 4). The probit function $\Phi : \mathbb{R} \to (0, 1)$ can be used as an observation model for the signs of the partial derivatives [31],

$$p(\boldsymbol{z}|f_F') = \prod_{(l,i)\in F} \Phi\left(z_{li}\cdot f_i'(t_l)\cdot rac{1}{v}
ight),$$

where f'_F denotes the partial derivative values $f'_i(t_l)$ in Equation (6) in the subset F of points. Function $\Phi(\cdot)$ is the standard Normal cumulative distribution function and v > 0 is a parameter controlling the strictness of the constraint. When v approaches zero ($v \rightarrow 0$), the function Φ approaches a step function. Setting $v = 10^{-4}$ seems a reasonable value for this study.

5.8. Likelihood Function

The joint likelihood of regular observations $y = (y_A, y_B)$ (Equations (9) and (8)), derivative value observations *m* (Equation (10)) and derivative sign observations *z* (Equation (11)), given the parameters β , *b* and σ is:

$$p(\boldsymbol{y}, \boldsymbol{m}, \boldsymbol{z} | \boldsymbol{\beta}, \boldsymbol{b}, \sigma) = p(\boldsymbol{y}_A | \boldsymbol{f}_A) \, p(\boldsymbol{y}_B | \boldsymbol{f}_B) \, p(\boldsymbol{z} | \boldsymbol{f}_F') \, p(\boldsymbol{m} | \boldsymbol{f}_E') = \left(\prod_{l,i \in B} \mathcal{N}(\boldsymbol{y}_{li} | \boldsymbol{f}_i(t_l), \sigma^2)\right) \left(\prod_{l,i \in B} \delta(\boldsymbol{y}_{li} - \boldsymbol{f}_i(t_l))\right) \left(\prod_{l,i \in F} \Phi(\boldsymbol{z}_{li} \cdot \boldsymbol{f}_i'(t_l) \cdot \frac{1}{v})\right) \left(\prod_{l,i \in E} \delta(\boldsymbol{m}_{li} - \boldsymbol{f}_i'(t_l))\right).$$
(12)

5.9. Posterior and Predictive Distributions

The posterior joint distribution of parameters given the data, which is proportional to the product of the likelihood and priors, is:

$$p(\boldsymbol{\beta}, \boldsymbol{b}, \sigma | \boldsymbol{y}, \boldsymbol{m}, \boldsymbol{z}) \propto p(\boldsymbol{y}, \boldsymbol{m}, \boldsymbol{z} | \boldsymbol{\beta}, \boldsymbol{b}, \sigma) p(\boldsymbol{\beta}) p(\boldsymbol{b} | \boldsymbol{\alpha}, \boldsymbol{\rho}) p(\boldsymbol{\alpha}) p(\boldsymbol{\rho}) p(\sigma),$$

where $p(\boldsymbol{y}, \boldsymbol{m}, \boldsymbol{z} | \boldsymbol{\beta}, \boldsymbol{b}, \sigma)$ is the likelihood of the model in Equation (12) and $p(\boldsymbol{b} | \boldsymbol{\alpha}, \boldsymbol{\rho}) = \prod_{k=1}^{K} \mathcal{N}(\boldsymbol{b}_k, | \boldsymbol{0}, \boldsymbol{\Sigma}_k)$ is the GP prior to the spline coefficients \boldsymbol{b} (Equation (5)). We set normal prior distributions for the linear coefficients $\boldsymbol{\beta}, p(\boldsymbol{\beta}) = \mathcal{N}(\boldsymbol{\alpha} | \boldsymbol{0}, 10I)$ and positive half-normal prior distributions for the hyperparameters $\boldsymbol{\alpha}, p(\boldsymbol{\alpha}) = \mathcal{N}^+(\boldsymbol{\alpha} | \boldsymbol{0}, 10)$ and $\sigma, p(\sigma) = \mathcal{N}^+(\sigma | \boldsymbol{0}, 10)$; and gamma distributions for the hyperparameters $\boldsymbol{\rho}$ and $p(\rho_d) = \text{Gamma}(\rho_d | 1, 0.1)$ for all d. These correspond to weakly informative prior distributions based on prior knowledge about the magnitude of the parameters.

The joint predictive distribution of new output values \tilde{y} and \tilde{y}' for a new sequence of input values x^* can be computed by integrating out over the posterior distribution:

$$p(\tilde{\boldsymbol{y}}, \tilde{\boldsymbol{m}}, \tilde{\boldsymbol{z}} | \boldsymbol{y}, \boldsymbol{m}, \boldsymbol{z}) = \int p(\tilde{\boldsymbol{y}}, \tilde{\boldsymbol{m}}, \tilde{\boldsymbol{z}} | \boldsymbol{\beta}, \boldsymbol{b}, \sigma) p(\boldsymbol{\beta}, \boldsymbol{b}, \sigma | \boldsymbol{y}, \boldsymbol{m}, \boldsymbol{z}) \, \mathrm{d}\boldsymbol{\beta} \, \mathrm{d}\boldsymbol{b} \, \mathrm{d}\sigma.$$

6. Bayesian Inference

The posterior distributions of interest $p(f, \sigma | y, m, z)$ and $p(\beta, b, \sigma | y, m, z)$ are in general not available in closed form. Hence, to estimate both the parameter posterior distribution and the posterior predictive distribution for this model, simulation methods and distributional [47] or numerical [48,49] approximation methods must be used. Simulation methods based on the Markov chain Monte Carlo (MCMC) [50], and more recently, the Hamiltonian Monte Carlo (HMC) [51], are general sampling methods to obtain samples from the joint posterior distribution. In this study, HMC methods are used to make inferences about the posterior and predictive distributions. Using sampling methods such as HMC, the covariance matrix must be inverted at every step of the sampler.

When computing the posterior distribution, the main computational demands come from the need of inverting the Gram matrix of the covariance function, usually through Cholesky factorization, of the hyperparameters in the GP prior defined over the spline coefficients b. The proposed model considers

the same prior spatial covariance structure for every vector of spline coefficients $\boldsymbol{b}_{.k}$ associated with a different order k of the spline functions. This requires $O(N^3)$ computations due to the covariance matrix inversion, where N denotes the number of observed spatial locations.

In general, the computational requirements to invert the covariance matrix in the proposed spatially correlated time series model framework is substantially reduced compared to using a full spatio-temporal GP model framework. The spatially correlated time series model framework, where a complete covariance structure among spline coefficients is considered, requires $O((NK)^3)$ operations in the covariance matrix inversion, whereas a spatio-temporal GP requires $O((NT)^3)$. Notice that the number of knots *K* in spline-based models is usually much lower than the number of time points *T* (i.e., $K \ll T$). Furthermore, it is meaningful, in the proposed splines-based model framework, to consider different covariance structures, with different complexities, between space and knots. In case a null covariance is considered between spline coefficients associated with different spline knots, the computational cost of the correlated spline model becomes $O(N^3K)$. Furthermore, if the same prior spatial structure is considered for the spline coefficients associated with different spline knots, as implemented in the current work, the computational cost of the proposed model becomes $O(N^3)$.

For a large *N* and *T* for which the covariance matrix inversion is computationally intractable, sparse [52] or basis functions [53] approximations for GPs can be used. A recently developed reduced-rank approximate GP based on Hilbert space methods was developed by Solin and Särkkä [38] and its practical implementation investigated by Riutort-Mayol, et al. [54]. For even larger datasets, where iterative simulation algorithms are too slow, modal and distributional approximation methods can be efficient alternatives.

7. Model Checking, Predictive Performance and Model Selection

Typical methods for checking normality and trends on the fitted residuals can be used to assess the model. Additionally, other criteria—in order to guarantee good model performance and ensure that the model is compatible with the data—are the posterior predictive checks, the expected log predictive density and the mean square predictive errors, which can be computed following cross-validation procedures.

The posterior predictive checks, which are also known as the leave-one-out probability integral transformation (LOO-PIT), can be used to assess whether the model predictive distributions are calibrated, that is, they are describing the model predictive uncertainty well. In case of a good calibration of predictive distribution, LOO-PIT values are uniformly distributed. They are based on computing the probability of a predictive value \tilde{y}_{il} to be lower than or equal to its corresponding actual observation y_{il} that has not taken part in fitting the model [47,55]:

LOO-PIT_{(l,i) \in \mathfrak{D}} =
$$P(\tilde{y}_{li} \le y_{li})$$
,

where \mathfrak{D} denotes the subset of observation indices of new predictions in the cross-validation; and $|\mathfrak{D}|$ is its cardinality; y_{li} denotes the actual observation at point (l, i) that does not take part in fitting the model; \tilde{y}_{li} denotes the predicted value from the model at point (l, i). Using simulation methods for inference, the calculation of the probability that a predicted value is less than observed in the LOO-PIT is straightforward through the collection of simulations.

From the models having well-calibrated predictive distributions, we would prefer the most certain one, which can be assessed using mean square predictive error (MSE) or even better using the expected log posterior predictive density (ELPD). The MSE evaluates, by averaging over all checking observations, how far new data are from the model by using the distance (error) between the actual observation y_i and the predictive mean \tilde{y}_{li} :

$$MSE = \frac{1}{|\mathfrak{D}|} \sum_{(l,i)\in\mathfrak{D}} (y_{li} - \tilde{y}_{li})^2.$$

Additionally, the ELPD evaluates, by averaging over all checking observations, how far new data are from the model while taking the posterior uncertainties into account. It is based on the log-density of new data given the model [56,57]:

$$\text{ELPD} = \frac{1}{|\mathfrak{D}|} \sum_{(l,i)\in\mathfrak{D}} \ln(p(y_{li}|\boldsymbol{y}_{-\mathfrak{D}})),$$

where $y_{-\mathfrak{D}}$ denotes the dataset without the subset \mathfrak{D} of observations, $y_{-\mathfrak{D}} = \{y_{li} : (l, i) \notin \mathfrak{D}\}.$

Following a leave one out cross-validation (denoted as loo-cv) scheme, the subset \mathfrak{D} of observation indices will be just a single observation (l, i), $\mathfrak{D} = \{(l, i)\}$. The LOO-PIT, ELPD, and MSE will be computed following the loo-cv scheme. The LOO-PIT will essentially be useful for model checking, assessing that the model is compatible with the data. The ELPD and MSE will evaluate the predictive accuracy of individual observations.

The end goal of this work is to predict complete color-fading time series at new unobserved locations. In order to do that, a leave one location out cross-validation (denoted as lolo-cv) scheme can be performed, where the subset of observation indices \mathfrak{D} will be a complete time series of a specific spatial location $i, \mathfrak{D} = \{(l, i) : l \in \{1, ..., T\}\}$. The statistics ELPD and MSE will be computed following the lolo-cv scheme. Plots of predicted new time series superimposed to their corresponding actual observations will be shown in order to visually evaluate the predictive performance. Model selection can be done by comparing the predictive performance between models using the ELPD and MSE statistics. The best model is the one that maximizes the ELPD and/or minimizes the MSE.

8. Experimental Results

The posterior and predictive distributions have been estimated using HMC sampling methods [51] using the Stan software [58]. Three simulation chains with different initial values have been launched. The convergence of the simulation chains was evaluated by the split-Rhat convergence diagnostic and the effective sample size of the chains [47,59]. A value of 1 in the split-Rhat convergence statistic indicates good mixing of simulated chains. Traditionally accepted good value for split-Rhat would be between 1 and 1.1. However, a more strict range has also been suggested [59] recently. In our case and for both models, a split-Rhat value lower than 1.05 has been obtained for all parameter simulation chains.

8.1. Posterior Distributions of Hyperparameters

Figure 3 depicts the estimated posterior marginal distributions of the hyperparameters: length-scales ρ associated with the input variables, overall magnitude α of the GP prior for the spline coefficients and observation noise σ of the model. Their mode estimates are shown in Table 2.



Figure 3. Posterior marginal density distributions of the hyperparameters: marginal variance α and length-scales $\rho = (\rho_1, \rho_2, \rho_3, \rho_4)$ of the GP prior for the spline coefficients, and residual noise σ of the model.

Input variables were previously re-scaled (standardized). The length-scale parameters control how the covariance function changes on the difference between the values of the different input variables

between locations. Large values indicate a slow decay of the covariance function as the difference between the input values increases and small values indicate a fast decay. The length-scale parameters ρ_1 and ρ_3 , corresponding to variables H and I, respectively, are relatively small. This indicates that the rate of decay of the correlation is moderately high. Therefore, variations in input variables Hand I imply a quick decrease in the correlations towards zero. Variables S, S_x , and S_y have larger length-scales. Notice that variables S_x and S_y share the same length-scale ($\rho_4 = \rho_5$), so that the covariance function depends on the Euclidean distance between spatial coordinates. This indicates that the covariance function depends on S and the Euclidean spatial distance in a smoother way.

Length-Scale of Spatial Input Variables				es	Marginal Magnitude of GP Prior for Coefficients b	Obs. Noise
Н	S	Ι	S_x	S_y		
ρ_1	ρ_2	$ ho_3$	$ ho_4$	$ ho_5$	α	σ
1.10	20.0	0.85	7	.9	0.58	0.38

Table 2. Estimated posterior modes of the hyperparameters.

8.2. Model Checking and Selection

Figure 4 shows the frequency histograms of the posterior predictive checks (LOO-PIT) by following the loo-cv scheme for both models, with and without derivatives. Both histograms are close to uniform distribution, which means the model predictions are well-calibrated.

Table 3 shows the ELPD and MSE statistics computed by following the two different cross-validation schemes, loo-cv and lolo-cv, for both models, with and without derivatives. The MSE for the model with derivatives is lower than the model without derivatives in the lolo-cv scheme. Furthermore, when the uncertainty is taken into account in the evaluation with the ELPD statistic, the improvement of using derivatives is even considerably larger in both cross-validation scenarios. Therefore, the results of these statistics confirm that the model with derivatives is closer to new data, either in terms of the expected log-density or the mean error.

		with Derivative Inform	ation without Derivative Information
loo-cv	ELPD MSE	-0.61 0.13	$\begin{array}{c} -0.78\\ 0.14\end{array}$
lolo-cv	ELPD MSE	-16.40 3.09	-33.39 4.42
	10 8 6 - 4 - 2 0 0	0 0.2 0.4 0.6 0.8 1.0 (a)	$ \begin{pmatrix} 10 \\ 8 \\ 6 \\ 4 \\ 2 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0$

Table 3. ELPD and MSE for both models, with and without derivative information, computed by the two cross-validation schemes, loo-cv and lolo-cv.

Figure 4. Histograms of the LOO probability integral transformation (LOO-PIT) by following loo-cv, for: (**a**) the model with derivatives, and (**b**) the model without derivatives.

8.3. Posterior and Predicted Time-Series Functions

Figure 5 shows the means and 95% pointwise credible intervals of predictive distributions, $p(\tilde{y}|y, m, z)$, evaluated over the sampling input space and plotted as a function of time (time series) for specific spatial locations and superimposed to the actual observations y. Additionally, the means of the predictive distribution of the model without the inclusion of the derivatives, $p(\tilde{y}|y)$, are also plotted for comparison.



Figure 5. Predictive distributions (means and 95% pointwise credible intervals) of the regular process with and without derivative information plotted as functions of time, and for specific sampling spatial locations. The actual data *y* have also been plotted as crosses.

The lolo-cv scheme, based on leaving a whole time series out of the training data, has been carried out in order to evaluate the prediction performance of complete new time series at new locations. Figure 6 shows the predicted time series at left-out locations (predictive distributions at left-out input locations and time points) following the lolo-cv scheme with derivative constraints, $p(\tilde{y}_{\cdot i}|y_{\cdot-i}, m_{\cdot-i}, z_{\cdot-i})$, and without them, $p(\tilde{y}_{\cdot i}|y_{\cdot-i})$, the consideration of derivatives. Here, $y_{\cdot-i}, m_{\cdot-i}$ and $z_{\cdot-i}$ denote the sets of observations y, m and z without those observations corresponding to time series i. The actual data of predicted time series $y_{\cdot i}$ in the loo-cv scheme are also plotted to visually evaluate the predictions. Closer predictions to the actual data, narrower predictive intervals (especially at the last part of the time series functions) and good shapes of the functions for the model

with derivatives can be appreciated, as monotonicity and saturation constraints improve and reduce the credible intervals of the predictions (Figure 6). The accuracy in mean and uncertainty of new predicted time series was evaluated by the MSE and ELPD in Table 3.



Figure 6. Predicted means and 95% pointwise credible intervals of new time series for predicting locations in the lolo-cv scheme, for both models, with and without derivatives. The actual data y have also been plotted as crosses.

It is worth mentioning that the use of an observation model with the absence of noise at the starting time points, $\{y_{li} = 0 : l = 1\}$, forces the predicted time series to be zero at those starting points (Figures 5 and 6).

Inducing monotonicity by means of virtual observations of the sign of the partial derivatives at two time points, $\{sign(\frac{\partial f_{li}}{\partial t_l}) = 1 : l \in \{7, 10\}\}$, has been sufficient to achieve monotonicity throughout the time series (Figures 5 and 6), preventing an overly smoothing effect on the posterior functions due to using many virtual points for monotonicity, as indicated in Riutort-Mayol [32], Chapter 4. In the same way, the consideration of additional observations of the value of the partial derivative equal to zero at the last time points, $\{\frac{\partial f_{li}}{\partial t_l} = 0 : l = 11\}$, induced a stationary state at the end of the time series.

Monotonicity and long term saturation properties of the curves were not assured when using the models without derivative information (Figures 5 and 6). Hence, the proposed model with derivative information yields a better fit and predictions for dynamics of the functions, improving their interpretability. In this sense, the analysis of the color fading curves using a model without derivative information could not be done properly because the temporal degradation, especially at the last time points, is unrealistic.

Prediction sensitivity due to the short set of data available has been found. Figure 6 shows poor predictive performance in some spatial locations when compared to the observed data. This is due to the high sensitivity of the model to leaving some data out since the dataset is scarce.

8.4. A Continuous Map of Color Fading Estimates

In order to make spatial continuous maps of color fading estimates, predictions of color fading time series, $p(\tilde{y}_{.j}|y, m, z)$, have been computed for all the spatial pixel locations *j* of the rock art painting image (Figure 2). In Figure 7 six images representing the spatial distributions and the evolutions over time of those color fading estimates are shown. The images correspond to the time points $t_3 = 3$, $t_4 = 4$, $t_5 = 5$, $t_6 = 6$ and $t_{11} = 11$, respectively.



Figure 7. Spatial distribution over a continuous area of the rock art painting's image and the temporal evolution at time points $t_3 = 3$, $t_4 = 4$, $t_5 = 5$, $t_6 = 6$ and $t_{11} = 11$, of mean color fading estimates.

The spatial distribution over time of color fading values seems to be quite unvarying. This was expected, since the time series adjusted in the different locations have similar patterns—smooth, monotonically increasing and tending to saturate in the long term, and the same prior spatial covariance

structure was specified through the time dimension. Color fading values, especially when they are low values, like in this case (the maximum is 7.64), are not worth being converted to the RGB color space and plotted as an image, because the color changes will be not visible in an RGB image. The best way is plotting color ΔE_{ab}^* values (Figure 7). The science of colorimetry argues that ΔE_{ab}^* values higher than approximately 3.5 would be perceptible for the human eye looking at the real object [39].

The actual equivalency of the time points variable used in microfading measurements in years depends on the hours and intensity of sunlight that affects the paintings on a changing daily basis. Without proper monitoring of light, this equivalency is difficult to obtain. Although this aspect of the research was not considered in the current study, future work will include an evaluation of the location and geographical orientation of the paintings together with long-term monitoring of light and UV radiation to estimate the dose acting upon the paintings in years.

9. Discussion

The preservation of rock art paintings has been an important research topic over the last decades. Research has shown that there is a need for developing and testing new methods for assessing the state of preservation of materials and their potential degradation due to exposure to environmental factors. Studies on rock art have traditionally focus on the one hand, on the documentation, understanding and interpretation of the themes depicted by ancient civilizations [60–63], and on the other hand, on the preservation of paintings and petroglyphs [2,64,65]. Due to the complexity of materials found in rock art paintings, a combination of analytical techniques is often needed to provide a comprehensive identification of the pigment, substrate systems at a particular site [66]. At the elemental analysis level, the use of scanning electron microscopy coupled with energy dispersive X-ray analysis (SEM-EDX) and X-ray fluorescence (XRF) spectrometry has been essential for enhancing the current understanding of these materials, as can be seen from various studies conducted on several rock art sites from Valencia, Spain [67-69]. In some cases, inductively coupled plasma-mass spectrometry (ICPMS) has been employed for elemental analysis due to its enhanced sensitivity and the possibility of conducting semi-quantitative analysis; this can be seen for example in two separate studies conducted in Spain [67] and Australia [70]. Most researchers agree that to provide a more accurate identification of the mineral content of rock art, molecular techniques must be used in conjunction with elemental analysis. For example, quantitative mineral analysis provided by X-ray diffractometry (XRD) has been used to differentiate the various forms of minerals. For example, Beck, et al. [71] employed XRD to differentiate crystalline structures of manganese oxide after conducting in situ analysis of rock art paintings in the cave of Rouffignac, France. Similarly, Aura Tortosa, et al. [67] identified the presence of either hematite or goethite in samples from Coves de Santa Maira in Spain. Moreover, advances in infrared and Raman spectroscopies have enhanced the possibilities of pigment and binder identification in rock art at the molecular level. A comprehensive review article by Bersani and Lottici [72] provides an overview of the use of Raman spectroscopy to identify pigments found in rock art sites around the globe. An example of the use of infrared spectroscopy to characterize binding media employed in Australian rock art can be found in the work by Horn [73].

Nevertheless, a multi-technique approach has been recognized as the most adequate as a single technique does not usually provides all the information required [74]. In addition to identifying the materials present in rock art, more recent studies have dealt with degradation aspects of materials. For example, studies have been conducted in sites in Argentina [75], Ethiopia [76], and France [77] using some of the techniques indicated above. Although it is generally believed that rock art pigments are not very light sensitive, some studies have demonstrated that exposure to sunlight can have damaging effects on them due to thermal and photochemical degradation, and spectral (color) changes of the materials are one of its main effects [4,9]. Studies dealing with the use of MFS testing to evaluate the photostability of pigments found in rock art are very limited. To the best of the authors' knowledge, there is only one similar microfading study available in which the author tested samples from the Mogao Grottoes in China [78]. MFS testing results are complementary to other chemical

characterization studies of pigments previously conducted on related sites [79] and to earlier integral documentation work using 2D and 3D digital techniques [62]. However, the objective of the present study was to evaluate the performance of the technique in situ to later generate a series of degradation maps by an innovative mathematical treatment of MFS curves in this field, which is based on a probabilistic model with correlation structure and the inclusion of constraints on the MFS curves to model their monotonic behavior and find their saturation plateau. A reliable and accurate extrapolation of the color change curves was carried out to estimate the potential color degradation that may occur simultaneously in different zones of the paintings with the aim of producing degradation maps that take into account the relative degradation factors calculated for the different colored areas. These degradation maps provide a novel insight in the documentation and analysis of potential degradation of the materials.

The color change estimates are based on standardized color spaces, as this permits the visualization of color/substrate systems over a period of time. These calculations over the microfading data extrapolation map can be used to produce digital renderings of problematic zones throughout the rock art paintings. The original images representing the current state of preservation of the rock art paintings are later retouched to simulate a future state using computer imaging software. These simulations can be used to identify areas more prone to degradation in the context of the whole object with the aim of creating and developing a preservation strategy for the paintings. During the development of such a strategy, several important parameters need to be taken into account, including the position of the sun at different times of the year and its connection with the irradiation of rock art paintings and sunlight levels reaching the shelter walls. The virtual images obtained through the proposed method are useful for communicating and discussing with cultural heritage professionals and site managers about the potential detrimental effects that sunlight can have on rock art paintings. These discussions are essential for developing a preservation strategy, which takes into consideration the acceptable level of change in these valuable objects. These actions have a direct impact on rock art tourism as they serve to increase the public awareness about human and natural degradation issues in prehistoric rock art sites.

The development of a preservation strategy is possible once the degree of damage caused by direct sunlight is estimated, understood and visualized. Preventive conservation measures aimed at minimizing the effects of direct sunlight on rock art paintings may include, for example, ways of minimizing and controlling total exposure on the paintings, preparing schedules for covering sensitive objects during the times of the year when the rate of potential damage is higher, and building artificial barriers and/or shelters in the form of walls (permanent or removable) or directly applying conservation materials to act as barriers between the sunlight and the painted surfaces [80]. For paintings exhibiting a very poor state of preservation, only remedial measures are possible, including either removal of endangered rock-paintings where protection cannot be guaranteed (non-recommended policy because it is against solid ICOMOS conservation and restauration doctrines) or creating virtual models of the paintings to be used and shared by museums (state-of-the-art trend).

Careful documentation, testing, mathematical modeling and visualization of long-term changes have significant practical implications for geoarchaeology, conservation, preservation and heritage studies. The results from the present study provide a valuable social resource to raise awareness about the importance of preserving rock art among stakeholders and members of the public, local communities and future generations. Archaeologists and cultural heritage documentation specialists were initially concerned with the use of MFS testing on actual rock art paintings [3]. However, the present study has provided answers to the questions formulated by researchers currently involved in the study of rock art paintings. Microfading is an adequate application tool to study the long-term behavior of rock/pigment systems exposed to sunlight radiation. For this reason, a multidisciplinary approach, which includes mathematical modeling of microfading data along with virtual visualization of paintings, can enhance the documentation of these sites by offering a new dimension in terms of ageing processes and photostability of their colored systems.

10. Conclusions

Color is an important aspect of the documentation and conservation of historical assets such as prehistoric rock art paintings. Therefore, the knowledge of the potential color degradation on these systems is crucial for eventual safeguarding and conservation. MFS measurements are difficult and lengthy to materialize, especially in these rock art systems, so an interpolation procedure to make accurate predictions for other locations on the surface in a scenario with a small sample dataset is important. Furthermore, these measurements in these systems show large fluctuations, so the consideration of constraints in the modeling to overcome possible modeling issues that may arise due to these large fluctuations is highly encouraged.

A reliable model for a dataset with small and large fluctuations based on spatially correlated time series has been formulated, in which the regular process is jointly modeled with the derivative process. Modeling the derivative process allows us to include model constraints related to the derivative of the functions in order to fit the desired properties of the color fading curves and minimize the effects of large fluctuations in the original observations. Furthermore, taking into account these first-order constraints has been demonstrated to be beneficial, either in terms of predictive performance or application-specific interpretability. Predictive capacity (as measured by MSE and ELPD) is considerably better with the model with derivative constraints compared to the model without them. However, high sensitivity of the models to leaving some data out has been found due to the small size of the data set.

GPs have been used to spatially correlate the spline time series by correlating their splines coefficients. A multivariate covariance function in a GP has allowed the usage of many predictors to evaluate the spatial covariance structure of the data. In this sense, the color space variables and the spatial distance in the covariance function have been included, and the colorimetric variables demonstrated being useful to correlate MFS data. The contribution of the spatial positions to this covariance structure has been found to matter; consequently, often and widely used traditional spatial or spatio-temporal models may detect a useful correlation structure in the data. Furthermore, GPs are one of the most natural ways to evaluate the covariance structure and make useful generalizations of the data, and this approach of combining spline functions for the temporal dimension and multivariate GPs to spatially correlate the spline coefficients is an original spatio-temporal approach, seldom implemented in practice.

Multivariate covariance metrics and zero and first-order constraints might be very hard to implement outside of a Bayesian framework and GP models. A GP is flexible enough and allowed us to properly model this complex covariance structure of the time series dependent on different input covariates. The Bayesian framework has allowed us to jointly use normally distributed observations with probit distributed observations of the sign of the partial derivatives, allowing us to fulfill the determinants of the behavior of the functions. A useful and comprehensive implementation of derivative constraints for spline functions and in a Bayesian framework has been illustrated in this work. Furthermore, the computational requirements of inverting the Gram matrix of the covariance function to compute the posterior distribution in the proposed spatially correlated time series model framework have been reduced substantially compared to using a spatio-temporal GP framework, since K (knots) $\ll T$ (time points). Finally, it is possible to consider other covariance structures for the relationship between space and knots. Even so, given the large number of points at which predictions must be made, the computational burden is large, in particular given that HMC algorithms are being used. To speed up computation, distributional or numerical approximation methods can be faster alternatives. In relation to this, a future line of research will be to use an approximate method for Bayesian inference that could speed up computation, making it easier to apply the methodology on several other rock art sites.

To conclude, a Bayesian statistical model with the considerations of constraints and multivariate GPs to spatially correlate the color fading curves is a significant advancement in the applied field, useful for making accurate predictions and ensuring good properties for the color fading estimates. Research shows that reliable evolution maps of color fading estimates can be elaborated by means of using the proposed model with derivative information in comparison to the model without derivative information. Furthermore, from a social and economic standpoint, these research results will help to improve the allocation of resources and conservation strategies by taking into account the possible degradation of this important and unique group of prehistoric rock art paintings over time. This will also be useful information, alone or in conjunction with the evaluations of other environmental damaging factors, with which to create better policies for heritage preservation and sustainable archaeological cultural tourism.

Author Contributions: Conceptualization, G.R.-M., J.L.L. and J.M.d.H.-M.; methodology, G.R.-M., V.G.-R., J.L.L. and J.M.d.H.-M.; software, G.R.-M.; validation, G.R.-M. and V.G.-R.; formal analysis, G.R.-M., V.G.-R., J.L.L. and J.M.d.H.-M.; investigation, G.R.-M.; resources, J.M.d.H.-M. and J.L.L.; data curation, G.R.-M. and J.M.d.H.-M.; writing—original draft preparation, G.R.-M.; writing—review and editing, V.G.-R., J.L.L., J.M.d.H.-M. and G.R.-M.; visualization, G.R.-M.; supervision, J.L.L., V.G.-R. and J.M.d.H.-M.; project administration, J.L.L.; funding acquisition, J.L.L., J.M.d.H.-M. and V.G.-R. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Spanish Ministerio de Economía y Competitividad projects HAR2014-59873-R and MTM2016-77501-P, Spanish Ministerio de Ciencia e Innovación grant PID2019-106341GB-I00, Consejería de Educación, Cultura y Deportes (JCCM) and FEDER grants PPIC-2014-001 and SBPLY/17/180501/000491.

Acknowledgments: The authors want to express their gratitude to the General Directorate of Culture and Heritage, Conselleria d'Educació, Investigació, Cultura i Esport, Generalitat Valenciana for giving us access to carry out research at the archaeological site.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Cassar, M.; Brimblecombe, P.; Nixon, T.; Price, C.; Sabbioni, C.; Saiz Jimenez, C.; Van Balen, K. Technological Requirements for Solutions in the Conservation and Protection of Historic Monuments and Archaeological Remains. Working Paper for the STOA Unit, 2001. Available online: https://discovery.ucl.ac.uk/id/eprint/ 13333 (accessed on 21 January 2020).
- Giesen, M.J.; Ung, A.; Warke, P.A.; Christgen, B.; Mazel, A.D.; Graham, D.W. Condition assessment and preservation of open-air rock art panels during environmental change. *J. Cult. Herit.* 2014, 15, 49–56. [CrossRef]
- del Hoyo-Meléndez, J.M.; Lerma, J.L.; López-Montalvo, E.; Villaverde, V. Documenting the light sensitivity of Spanish Levantine rock art paintings. *Isprs Ann. Photogramm. Remote. Sens. Spat. Inf. Sci.* 2015, 2, 53–59. [CrossRef]
- 4. Díez-Herrero, A.; Gutiérrez-Pérez, I.; Lario, J.; Cañaveras, J.C.; Benavente, D.; Sánchez-Moral, S.; Alonso-Azcárate, J. Analysis of potential direct insolation as a degradation factor of cave paintings in Villar del Humo, Cuenca, Central Spain. *Geoarchaeol. Int. J.* **2009**, *24*, 450–465. [CrossRef]
- 5. del Hoyo-Meléndez, J.M.; Carrión-Ruiz, B.; Riutort-Mayol, G.; Lerma, J.L. Lightfastness assessment of Levantine rock art by means of microfading spectrometry. *Color Res. Appl.* **2019**, *44*, 547–555. [CrossRef]
- 6. Domingo Sanz, I.; Vendrell, M.; Chieli, A. A critical assessment of the potential and limitations of physicochemical analysis to advance knowledge on Levantine rock art. *Quat. Int.* **2020**. [CrossRef]
- 7. Hall, K.; Meiklejohn, I.; Sumner, P.; Arocena, J. Light penetration into Clarens sandstone and implications for deterioration of San rock art. *Geoarchaeol. Int. J.* **2010**, *25*, 122–136. [CrossRef]
- Mol, L.; Viles, H. Geoelectric investigations into sandstone moisture regimes: Implications for rock weathering and the deterioration of San Rock Art in the Golden Gate Reserve, South Africa. *Geomorphology* 2010, 118, 280–287. [CrossRef]

- 9. Hall, K.; Meiklejohn, I.; Arocena, J. The thermal responses of rock art pigments: Implications for rock art weathering in southern Africa. *Geomorphology* **2007**, *91*, 132–145. [CrossRef]
- 10. Feller, R.L.; Johnston-Feller, R.M.; Bailie, C. Determination of the specific rate constant for the loss of a yellow intermediate during the fading of alizarin lake. *J. Am. Inst. Conserv.* **1986**, *25*, 65–72. [CrossRef]
- 11. Giles, C.H. The fading of colouring matters. J. Appl. Chem. 1965, 15, 541–550. [CrossRef]
- Giles, C.H.; Johari, D.P.; Shah, C.D. Some observations on the kinetics of dye fading. *Text. Res. J.* 1968, 38, 1048–1056. [CrossRef]
- 13. Johnston-Feller, R.; Feller, R.L.; Bailie, C.W.; Curran, M. The kinetics of fading: opaque paint films pigmented with alizarin lake and titanium dioxide. *J. Am. Inst. Conserv.* **1984**, 23, 114–129. [CrossRef]
- 14. Whitmore, P.M.; Pan, X.; Bailie, C. Predicting the fading of objects: Identification of fugitive colorants through direct nondestructive lightfastness measurements. *J. Am. Inst. Conserv.* **1999**, *38*, 395–409. [CrossRef]
- 15. Whitmore, P.M.; Bailie, C.; Connors, S.A. Micro-fading tests to predict the result of exhibition: progress and prospects. *Stud. Conserv.* **2000**, *45*, 200–205. [CrossRef]
- 16. del Hoyo-Meléndez, J.M.; Mecklenburg, M.F. A survey on the light-fastness properties of organic-based Alaska Native artifacts. *J. Cult. Herit.* **2010**, *11*, 493–499. [CrossRef]
- 17. Wood, S.N. Thin plate regression splines. J. R. Stat. Soc. Ser. Stat. Methodol. 2003, 65, 95–114. [CrossRef]
- 18. Currie, I.D.; Durban, M.; Eilers, P.H. Generalized linear array models with applications to multidimensional smoothing. *J. R. Stat. Soc. Ser. Stat. Methodol.* **2006**, *68*, 259–280. [CrossRef]
- Lee, D.J.; Durbán, M. P-spline ANOVA-type interaction models for spatio-temporal smoothing. *Stat. Model.* 2011, 11, 49–69. [CrossRef]
- 20. Kneib, T.; Fahrmeir, L. Structured additive regression for categorical space–time data: A mixed model approach. *Biometrics* **2006**, *62*, 109–118. [CrossRef]
- 21. Aguilera-Morillo, M.C.; Durbán, M.; Aguilera, A.M. Prediction of functional data with spatial dependence: a penalized approach. *Stoch. Environ. Res. Risk Assess.* **2017**, *31*, 7–22. [CrossRef]
- 22. Neal, R. Regression and classification using Gaussian process priors (with discussion). In *Bayesian Statistics*; Bernardo, J.M., Berger, J.O., Dawid, A.P., Smith, A.F.M., Eds.; Oxford University Press: Oxford, UK, 1999; pp. 475–501.
- 23. Rasmussen, C.E.; Williams, C.K. *Gaussian Processes for Machine Learning*; The MIT Press: Cambridge, MA, USA, 2006.
- 24. Banerjee, S.; Carlin, B.P.; Gelfand, A.E. *Hierarchical Modeling and Analysis for Spatial Data*; Chapman and Hall/CRC: Boca Raton, FL, USA, 2014.
- 25. Cressie, N.; Huang, H.C. Classes of nonseparable, spatio-temporal stationary covariance functions. *J. Am. Stat. Assoc.* **1999**, *94*, 1330–1339. [CrossRef]
- 26. De Iaco, S.; Myers, D.E.; Posa, D. Nonseparable space-time covariance models: some parametric families. *Math. Geol.* **2002**, *34*, 23–42. [CrossRef]
- 27. Giraldo, R.; Delicado, P.; Mateu, J. Continuous time-varying kriging for spatial prediction of functional data: An environmental application. *J. Agric. Biol. Environ. Stat.* **2010**, *15*, 66–82. [CrossRef]
- Baladandayuthapani, V.; Mallick, B.K.; Young Hong, M.; Lupton, J.R.; Turner, N.D.; Carroll, R.J. Bayesian hierarchical spatially correlated functional data analysis with application to colon carcinogenesis. *Biometrics* 2008, 64, 64–73. [CrossRef]
- 29. Rasmussen, C.E. Gaussian Processes to speed up Hybrid Monte Carlo for expensive Bayesian integrals. In *Bayesian Statistics*; Oxford University Press: Oxford, UK, 2003; pp. 651–659.
- 30. Shively, T.S.; Sager, T.W.; Walker, S.G. A Bayesian approach to non-parametric monotone function estimation. *J. R. Stat. Soc. Ser. Stat. Methodol.* **2009**, *71*, 159–175. [CrossRef]
- 31. Riihimäki, J.; Vehtari, A. Gaussian processes with monotonicity information. In Proceedings of the Thirteenth International Conference on Artificial Intelligence and Statistics, AISTATS 2010, Chia Laguna Resort, Sardinia, Italy, 13–15 May 2010; pp. 645–652.
- 32. Riutort-Mayol, G. Approximate Gaussian Processes and Derivative Information for Spatio-Temporal Regression and Classification. Ph.D. Thesis, Universitat de València, Valencia, Spain, 2020.
- 33. Lorenzi, M.; Filippone, M. Constraining the Dynamics of Deep Probabilistic Models. 2018. Available online: https://arxiv.org/abs/1802.05680 (accessed on 13 December 2018).

- 34. Ramsay, J.O. Estimating smooth monotone functions. J. R. Stat. Soc. Ser. Stat. Methodol. 1998, 60, 365–375. [CrossRef]
- 35. Brezger, A.; Steiner, W.J. Monotonic regression based on bayesian p–splines: An application to estimating price response functions from store-level scanner data. *J. Bus. Econ. Stat.* **2008**, *26*, 90–104. [CrossRef]
- 36. Reich, B.J.; Fuentes, M.; Dunson, D.B. Bayesian spatial quantile regression. *J. Am. Stat. Assoc.* **2011**, 106, 6–20. [CrossRef]
- Andersen, M.R.; Siivola, E.; Riutort-Mayol, G.; Vehtari, A. A Non-Parametric Probabilistic Model for Monotonic Functions. 2018. Available online: https://sites.google.com/view/nipsbnp2018/acceptedpapers (accessed on 12 May 2019).
- Solin, A.; Särkkä, S. Hilbert space methods for reduced-rank Gaussian process regression. *Stat. Comput.* 2020, 30, 419–446. [CrossRef]
- 39. Malacara, D. *Color Vision and Colorimetry: Theory and Applications;* SPIE Press: Washington, DC, USA, 2011; Volume PM204.
- 40. Ford, B. Non-destructive microfade testing at the National Museum of Australia. *AICCM Bull.* **2011**, 32, 54–64. [CrossRef]
- 41. Ford, B.; Druzik, J. Microfading: the state of the art for natural history collections. *Collect. Forum.* **2013**, *27*, 54–57.
- 42. Columbia, M.R.; Smith, G.D.; Hoevel, C.; Messier, P. The Application of Microfadeometric Testing to Mounted Photographs at the Indianapolis Museum of Art. *Microsc. Microanal.* **2013**, *19*, 1412–1413. [CrossRef]
- 43. Crainiceanu, C.; Ruppert, D.; Wand, M.P. Bayesian Analysis for Penalized Spline Regression Using WinBUGS. *J. Stat. Softw.* **2005**, *14*, 1–24. [CrossRef]
- 44. Ruppert, D.; Wand, M.P.; Carroll, R.J. *Semiparametric Regression*; Number 12; Cambridge University Press: Cambridge, MA, USA, 2003.
- 45. Nievergelt, Y. Splines in Single and Multivariable Calculus. UMAP Module 1993, 718, 39–101.
- Brumback, B.A.; Ruppert, D.; Wand, M.P. Variable selection and function estimation in additive nonparametric regression using a data-based prior: Comment. J. Am. Stat. Assoc. 1999, 94, 794–797. [CrossRef]
- 47. Gelman, A.; Carlin, J.B.; Stern, H.S.; Dunson, D.B.; Vehtari, A.; Rubin, D.B. *Bayesian Data Analysis*, 3rd ed.; CRC Press: Boca Raton, FL, USA, 2013.
- 48. Rue, H.; Martino, S.; Chopin, N. Approximate Bayesian inference for latent Gaussian models by using integrated nested Laplace approximations. *J. R. Stat. Soc. Ser. B Stat. Methodol.* **2009**, *71*, 319–392. [CrossRef]
- 49. Gómez-Rubio, V. Bayesian Inference with INLA; CRC Press: Boca Raton, FL, USA, 2020.
- 50. Brooks, S.; Gelman, A.; Jones, G.; Meng, X.L. *Handbook of Markov Chain Monte Carlo*; CRC Press: Boca Raton, FL, USA, 2011.
- 51. Neal, R.M. MCMC using Hamiltonian dynamics. *In Handbook of Markov Chain Monte Carlo*; Brooks, S., Gelman, A., Jones, G.L., Meng, X.L., Eds.; CRC Press: Boca Raton, FL, USA, 2011; pp. 113–162.
- 52. Quiñonero-Candela, J.; Rasmussen, C.E. A unifying view of sparse approximate Gaussian process regression. *J. Mach. Learn. Res.* **2005**, *6*, 1939–1959.
- 53. Cramér, H.; Leadbetter, M.R. *Stationary and Related Stochastic Processes: Sample Function Properties and Their Applications*; Courier Corporation: New York, NY, USA, 2013.
- Riutort-Mayol, G.; Bürkner, P.C.; Andersen, M.R.; Solin, A.; Vehtari, A. Practical Hilbert Space Approximate Bayesian Gaussian Processes for Probabilistic Programming. 2020. Available online: https://arxiv.org/abs/ 2004.11408 (accessed on 23 April 2020).
- 55. Gelfand, A.E.; Dey, D.K.; Chang, H. *Model Determination Using Predictive Distributions with Implementation via Sampling-Based Methods*; Technical Report; Stanford University CA Department of Statistics: Stanford, CA, USA, 1992.
- 56. Vehtari, A.; Ojanen, J. A survey of Bayesian predictive methods for model assessment, selection and comparison. *Stat. Surv.* **2012**, *6*, 142–228. [CrossRef]
- Andersen, M.R.; Magnusson, M.; Jonasson, J.; Vehtari, A. Bayesian Leave-One-Out Cross-Validation for Large Data. In Proceedings of the 36th International Conference on Machine Learning, Long Beach, CA, USA, 10–15 June 2019.

- Carpenter, B.; Gelman, A.; Hoffman, M.D.; Lee, D.; Goodrich, B.; Betancourt, M.; Brubaker, M.; Guo, J.; Li, P.; Riddell, A. Stan: A probabilistic programming language. J. Stat. Softw. 2017, 76, 1649–1660. [CrossRef]
- 59. Vehtari, A.; Gelman, A.; Simpson, D.; Carpenter, B.; Bürkner, P.C. Rank-normalization, folding, and localization: An improved \hat{R} for assessing convergence of MCMC. *arXiv* **2019**, arXiv:1903.08008.
- 60. Whitley, D.S. Notes on the Coso petroglyphs, the etiological mythology of the western Shoshone, and the interpretation of rock art. *J. Calif. Great Basin Anthropol.* **1982**, *4*, 262–272.
- 61. Manhire, T.; Parkington, J.; Van Rijssen, B. A distributional approach to the interpretation of rock art in the south-western Cape. *Goodwin Ser.* **1983**, *6*, 29–33. [CrossRef]
- 62. Domingo, I.; Villaverde, V.; López-Montalvo, E.; Lerma, J.L.; Cabrelles, M. Latest developments in rock art recording: towards an integral documentation of Levantine rock art sites combining 2D and 3D recording techniques. *J. Archaeol. Sci.* **2013**, *40*, 1879–1889. [CrossRef]
- 63. Rissolo, D. In the Realm of Rain Gods: A Contextual Survey of Rock Art across the Northern Maya Lowlands. *Heritage* **2020**, *3*, 1094–1108. [CrossRef]
- 64. Howard, A.J. Managing global heritage in the face of future climate change: The importance of understanding geological and geomorphological processes and hazards. *Int. J. Herit. Stud.* **2013**, *19*, 632–658. [CrossRef]
- Groom, K.M.; Villa Cerveny, N.; Allen, C.D.; Dorn, R.I.; Theuer, J. Protecting Stone Heritage in the Painted Desert: Employing the Rock Art Stability Index in the Petrified Forest National Park, Arizona. *Heritage* 2019, 2, 2111–2123. [CrossRef]
- Chalmin, E.; Menu, M.; Vignaud, C. Analysis of rock art painting and technology of Palaeolithic painters. *Meas. Sci. Technol.* 2003, 14, 1590. [CrossRef]
- Aura Tortosa, J.E.; Gallello, G.; Roldán, C.; Cavallo, G.; Pastor, A.; Murcia-Mascarós, S. Characterization and sources of Paleolithic–Mesolithic ochre from Coves de Santa Maira (Valencian Region, Spain). *Geoarchaeology* 2020, 1–20. [CrossRef]
- López-Montalvo, E.; Roldán, C.; Badal, E.; Murcia-Mascarós, S.; Villaverde, V. Identification of plant cells in black pigments of prehistoric Spanish Levantine rock art by means of a multi-analytical approach. A new method for social identity materialization using chaîne opératoire. *PLoS ONE* 2017, 12, e0172225. [CrossRef]
- 69. López-Montalvo, E.; Villaverde, V.; Roldán, C.; Murcia, S.; Badal, E. An approximation to the study of black pigments in Cova Remigia (Castellón, Spain). Technical and cultural assessments of the use of carbon-based black pigments in Spanish Levantine Rock Art. *J. Archaeol. Sci.* **2014**, *52*, 535–545. [CrossRef]
- 70. Scadding, R.; Winton, V.; Brown, V. An LA-ICP-MS trace element classification of ochres in the Weld Range environ, Mid West region, Western Australia. *J. Archaeol. Sci.* **2015**, *54*, 300–312. [CrossRef]
- Beck, L.; Rousseliere, H.; Castaing, J.; Duran, A.; Lebon, M.; Moignard, B.; Plassard, F. First use of portable system coupling X-ray diffraction and X-ray fluorescence for in-situ analysis of prehistoric rock art. *Talanta* 2014, 129, 459–464. [CrossRef]
- 72. Bersani, D.; Lottici, P. Raman spectroscopy of minerals and mineral pigments in archaeometry. *J. Raman Spectrosc.* **2016**, *47*, 499–530. [CrossRef]
- 73. Horn, K.R. Time takes its toll: Detection of organic binder media in ochre paints with visible near-infrared and short-wave infrared reflectance spectroscopy. *J. Archaeol. Sci. Rep.* **2018**, *21*, 10–20. [CrossRef]
- 74. Stuart, B.; Thomas, P. Pigment characterisation in Australian rock art: A review of modern instrumental methods of analysis. *Herit. Sci.* **2017**, *5*, 10. [CrossRef]
- 75. Rousaki, A.; Vargas, E.; Vázquez, C.; Aldazábal, V.; Bellelli, C.; Calatayud, M.C.; Hajduk, A.; Palacios, O.; Moens, L.; Vandenabeele, P. On-field Raman spectroscopy of Patagonian prehistoric rock art: Pigments, alteration products and substrata. *Trac. Trends Anal. Chem.* **2018**, *105*, 338–351. [CrossRef]
- 76. Gallinaro, M.; Zerboni, A. Rock, pigments, and weathering. A preliminary assessment of the challenges and potential of physical and biochemical studies on rock art from southern Ethiopia. *Quat. Int.* **2020**. [CrossRef]
- 77. Lacanette, D.; Large, D.; Ferrier, C.; Aujoulat, N.; Bastian, F.; Denis, A.; Jurado, V.; Kervazo, B.; Konik, S.; Lastennet, R.; et al. A laboratory cave for the study of wall degradation in rock art caves: An implementation in the Vézère area. *J. Archaeol. Sci.* **2013**, *40*, 894–903. [CrossRef]
- 78. Druzik, J.R. Evaluating the light sensitivity of paints in selected wall paintings at the Mogao Grottoes: Caves 217, 98, and 85. In *Conservation of Ancient Sites on the Silk Road: Proceedings of the Second International Conference on the Conservation of Grotto Sites, Mogao Grottoes, Dunhuang, China, 28 June 28–3 July 2004;* Getty Publications: Los Angeles, CA, USA, 2010; pp. 457–463.

- Roldán, C.; Murcia-Mascarós, S.; Ferrero, J.; Villaverde, V.; López, E.; Domingo, I.; Martínez, R.; Guillem, P.M. Application of field portable EDXRF spectrometry to analysis of pigments of Levantine rock art. *X-ray Spectrom. Int. J.* 2010, *39*, 243–250. [CrossRef]
- 80. Avery, G. The preservation of rock-art with special reference to South African problems and conditions. *S. Afr. Archaeol. Bull.* **1975**, *30*, 139–142. [CrossRef]

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).